Author's response to reviews

Title: Transfusion Related Acute Lung Injury with massive pulmonary secretion during cardiac surgery. A case report.

Authors:

Julien Teodori (julien.teodori@gmail.com)
Kamal Rampersad (karampersad@yahoo.com)
Giovanni Teodori (giovanni.teodori1@gmail.com)
Roland Roopchand (rolandroopchand@yahoo.com)
Gianni Angelini (g.d.angelini@imperial.ac.uk)

Version: 3 Date: 13 February 2014

Author's response to reviews: see over
Dear Editor of Journal of Cardiothoracic Surgery

I would be grateful if you would consider the enclosed case report:


Authors: J. Teodori, MD, K. Rampersad, MD, G. Teodori, MD, R. Roopchand, MD and G. D.Angelini, FRCS*.

Caribbean Heart Care Medcorp, Port Of Spain, Trinidad and Tobago; * NILH Imperial College London, London, UK.

Corresponding author: G. D.Angelini, FRCS

Email addresses; Julien Teodori: julien.teodori@gmail.com; Giovanni Teodori: giovanni.teodori1@gmail.com; Kamal Rampersad: karampersad@yahoo.com; Roland Roopchand: rolandroopchand@yahoo.com; Gianni Angelini: g.d.angelini@imperial.ac.uk

Possible reviewers are:
Thomas A. Salerno tsalerno@med.miami.edu
Antonio Maria Calafiore am.calafiore@gmail.com

Regards,

Julien Teodori M.D.

Port of Spain, Trinidad, the 10th Nov. 2013